Author's response to reviews

Title: The impact of generic-only drug benefits on patients' use of inhaled corticosteroids in a Medicare population with asthma

Authors:

Vicki Fung (Vicki.Fung@kp.org)
Ira B Tager (ibt@berkeley.edu)
Richard Brand (ribrand@sbcglobal.net)
Joseph P Newhouse (newhouse@hcp.med.harvard.edu)
John Hsu (jth@dor.kaiser.org)

Version: 6 Date: 15 July 2008

Author's response to reviews:

Dear Editors,

I have revised the manuscript to correct for a few minor typographical errors.

Best,

Vicki Fung